Author's response to reviews

Title: Apple Phytochemicals and Their Health Benefits

Authors:

   Prof Jeanelle Boyer (rl23@cornell.edu)  
   Prof Rui Hai Liu (RL23@cornell.edu)

Version: 2 Date: 5 Mar 2004

PDF covering letter
January 19, 2004

Dr Nehme Gebrayel, MD
Editor-in-Chief
Leesburg Regional Medical Center
Leesburg
Florida
United States

Dear Dr. Gebrayel:

Thank you for your e-mail encouraging me to submit our review to be considered for publication in Nutritional Journal. Please find the attached manuscript titled “Apple Phytochemicals and Their Health Benefits” by Boyer and Liu. We would like to have this manuscript considered for publication in an upcoming issue of Nutritional Journal. This is original work and not under consideration elsewhere.

Potential reviewers include:

1. Professor David Jacobs
   University of Minnesota/School of Public Health, Suite 300, Minneapolis, MN 55454
   Tel. 612-624-4196, Fax 612-624-0315, E-mail: Jacobs@epi.umn.edu

2. Dr. Mary Ellen Camire
   Associate Professor
   Department of Food Science and Human Nutrition, The University of Maine, 5736 Holmes Hall, Room 101, Orono, ME 04469-5736
   Tel. (207) 581-1627, Fax: (207) 581-1636; E-mail: camire@maine.edu

3. Professor Elizabeth H. Jeffery, Ph.D.
   Department of Food Science and Human Nutrition
   College of Agricultural, Consumer and Environmental Sciences
   1101 West Peabody Drive
   Urbana, Illinois 61801
   Tel. (217) 333-3820, Fax (217) 333-8046, E-mail: ejeffery@uiuc.edu

4. Dr. Youling Xiong
   Professor of Food Chemistry
   Department of Animal Science, University of Kentucky, 206 W.P. Garrigus Bldg., Lexington, KY 40546
   Tel. (859) 257-3822, Fax (859) 257-5318, E-mail: ylxiong@pop.uky.edu
5. Professor Joan Sabate  
Department of Nutrition  
School of Public Health  
Loma Linda University  
Loma Linda, California 92350  
Tel. (909) 558-4598, Fax (909) 558-4095, E-mail: jsabate@sph.llu.edu

Thank you in advance for your consideration. Please direct all correspondence directly to me at the above address.

Sincerely,

Rui Hai Liu, M.D., Ph.D.